

OPEN ACCESS

Citation: Wilke ABB, Mhlanga A, Kummer AG, Vasquez C, Moreno M, Petrie WD, et al. (2023) Diel activity patterns of vector mosquito species in the urban environment: Implications for vector control strategies. PLoS Negl Trop Dis 17(1): e0011074. https://doi.org/10.1371/journal.pntd.0011074

Editor: Gordana Rasic, QIMR Berghofer Medical Research Institute, AUSTRALIA

Received: September 7, 2022

Accepted: January 6, 2023

Published: January 26, 2023

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All necessary data to replicate the experiments done in this study have been made available within the manuscript.

Funding: The authors received no specific funding for this work.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: M.A. has received research funding from Seqirus; the funding is not related to this study. The other authors declare no competing interests.

RESEARCH ARTICLE

Diel activity patterns of vector mosquito species in the urban environment: Implications for vector control strategies

André B. B. Wilke¹*, Adequate Mhlanga¹, Allisandra G. Kummer¹, Chalmers Vasquez², Maday Moreno², William D. Petrie², Art Rodriguez³, Christopher Vitek⁴, Gabriel L. Hamer⁵, John-Paul Mutebi⁶, Marco Ajelli¹*

 Laboratory for Computational Epidemiology and Public Health, Department of Epidemiology and Biostatistics, Indiana University School of Public Health, Bloomington, Indiana, United States of America,
 Miami-Dade County Mosquito Control Division, Miami, Florida, United States of America,
 Public Health Department, City of Brownsville, Brownsville, Texas, United States of America,
 Center for Vector-Borne Diseases, The University of Texas Rio Grande Valley, Texas, United States of America,
 Department of Entomology, Texas A&M University, College Station, Texas, United States of America,
 Arboviral Diseases Branch (ADB), Division of Vector-Borne Diseases (DVBD), Centers for Disease Control and Prevention (CDC), Fort Collins, Colorado, United States of America

* andwilke@iu.edu (A.B.B.W.); marco.ajelli@gmail.com (M.A.)

Abstract

Mathematical models have been widely used to study the population dynamics of mosquitoes as well as to test and validate the effectiveness of arbovirus outbreak responses and mosquito control strategies. The objective of this study is to assess the diel activity of mosguitoes in Miami-Dade, Florida, and Brownsville, Texas, the most affected areas during the Zika outbreak in 2016–2017, and to evaluate the effectiveness of simulated adulticide treatments on local mosquito populations. To assess variations in the diel activity patterns, mosquitoes were collected hourly for 96 hours once a month from May through November 2019 in Miami-Dade County, Florida, and Brownsville, Texas. We then performed a PERMA-NOVA followed by a SIMPER analysis to assess whether the abundance and species richness significantly varies at different hours of the day. Finally, we used a mathematical model to simulate the population dynamics of 5 mosquito vector species and evaluate the effectiveness of the simulated adulticide applications. A total of 14,502 mosquitoes comprising 17 species were collected in Brownsville and 10,948 mosquitoes comprising 19 species were collected in Miami-Dade County. Aedes aegypti was the most common mosquito species collected every hour in both cities and peaking in abundance in the morning and the evening. Our modeling results indicate that the effectiveness of adulticide applications varied greatly depending on the hour of the treatment. In both study locations, 9 PM was the best time for adulticide applications targeting all mosquito vector species; mornings/afternoons (9 AM- 5 PM) yielded low effectiveness, especially for Culex species, while at night (12 AM-6 AM) the effectiveness was particularly low for Aedes species. Our results indicate that the timing of adulticide spraying interventions should be carefully considered by local authorities based on the ecology of the target mosquito species in the focus area.

Author summary

Mathematical models have been widely used to study vector mosquitoes as well as to test the effectiveness of arbovirus outbreak response and mosquito control strategies. However, due to the lack of empirical data, there are no studies focusing on the effectiveness of adulticide applications at different hours of the day to control different mosquito populations. This study leveraged a unique dataset in which approximately 25,000 mosquitoes comprising 19 species were collected hourly in Miami-Dade County, Florida, and Brownsville, Texas. We then developed a mathematical model to simulate the population dynamics of five mosquito vector species to evaluate the effectiveness of adulticide spraying at different times of the day and at different frequencies. Mosquito community composition and abundance varied significantly throughout the day in both Brownsville and Miami-Dade County with more than 10-fold differences during the day. Depending on the target vector species, the application of adulticides at a given hour of the day may lead to drastically different results, although we found some common patterns such as the remarkable effectiveness of interventions performed at 9 PM.

Introduction

The invasion and proliferation of mosquito vector species across new areas directly affect the epidemiology of mosquito-borne diseases, especially in complex socio-ecological urban ecosystems [1-3]. Due to the increase in the presence and abundance of vector mosquito species, major cities in the contiguous United States have been affected by local arbovirus transmission [4,5]. In particular, southern Florida and southern Texas have historically been afflicted by arbovirus outbreaks. They were the most affected areas of the United States during the Zika virus outbreak of 2016 [3,6] and have remained the theater of locally transmitted arboviral infections ever since [5,7,8].

The constant increase in the number of arboviral infections in historically unaffected areas is most likely due to the increase in the contact rate between mosquito vector species and human hosts [9,10]. Controlling mosquitoes is considered the most effective way to prevent arbovirus outbreaks; however, controlling mosquitoes in urban areas is a difficult task [11]. Large mosquito populations, including vector species, are now harbored in urban areas thanks to an overabundance of resources (including widely available artificial aquatic habitats for larval development and human hosts for blood-feeding), lack of or reduced density of natural predators, and reduced larval competition [12–14].

One of the most common strategies to control mosquitoes in urban areas is to reduce the number of active host-seeking females by spraying adulticides [15], especially in non-emergency situations, and in developing countries with limited access to more expensive and ecologically friendly alternatives such as bacterial larvicides and insect growth regulators. One of the most important elements that separate successful from unsuccessful adulticide spraying is the extent to which the target adult mosquito population is exposed to the adulticide [16]. To effectively reduce a mosquito population, adulticide applications must coincide with the period when the mosquitoes are most active (actively flying). On the other hand, adulticide interventions will have limited effects on resting adult mosquitoes. Therefore, knowing the period(s) of the day mosquitoes are most active is of paramount importance for vector control operations based on adulticide treatments and for the prevention and control of local arboviral transmission. To this aim, we collected mosquitoes hourly for 96 hours once a month for seven months in Miami-Dade County, Florida, and Brownsville, Texas. We then developed a mathematical model informed with the collected data to simulate "in silico" the population dynamics of 5 mosquito vector species and the potential effect of adulticide applications at different times of the day and frequencies.

Methods

Mosquito collection and study sites

Mosquitoes were collected by BG-Sentinel 2 traps (Biogents AG, Regensburg, Germany) baited with BG Lures and dry ice as a source of carbon dioxide in Miami-Dade County, Florida, and Brownsville, Texas, as described in Mutebi et al. [17]. The study locations were chosen due to their history of arboviral outbreaks, including West Nile, yellow fever, dengue, and Zika viruses. In each city, four collection sites were selected to capture the natural variation in the diel activity patterns of mosquito species due to their inherent socio-ecological features. BG-Sentinel 2 traps are the gold standard to assess the community composition of mosquitoes in urban environments. They mimic a potential host by using BG-Lures that release attractants consisting of carbon dioxide, lactic acid, ammonia, and caproic acid, substances that are found on animal skin thus attracting female mosquitoes seeking blood sources. In addition, the BG traps were baited with dry ice to release carbon dioxide, which has been shown to significantly increase the capture of diverse mosquito species [18]. Male mosquitoes collected during this study were considered accidental catches and, since they were not consistently collected, they were not considered informative and were not included in the analyses.

Mosquito collections were made from May 2019 to November 2019 as previously described by Mutebi et. al. [17]. Mosquitoes were collected during the warmer months of the year (at the study sites) to maximize the number of collectable specimens. Each BG-Sentinel trap was monitored every hour for 96 hours at each location once every month from May to November 2019. Every hour, the BG-Sentinel traps were serviced by project personnel, and the collection bags were removed from the traps and replaced with empty ones. The bags of mosquitoes were then labeled and transported to the laboratory under refrigeration, where they were subsequently identified to species on chill tables by using entomological keys [19].

Diversity analysis

To determine the extent to which the diel activity patterns of each species as well as the abundance and species richness varied in each hour of the day, and if these differences were statistically significant, we assessed the hourly community composition, richness, and abundance of mosquitoes. To assess variations in the diel activity patterns of mosquito species in each city, we organized the data into six groups (group 1 = 12 AM to 3 AM; group 2 = 4 AM to 7 AM; group 3 = 8 AM to 11 AM; group 4 = 12 PM to 3 PM; group 5 = 4 PM to 7 PM; and group 6 = 8 PM to 11 PM) and performed a PERMANOVA with 9,999 permutations based on Bray-Curtis (abundance-based) and the Jaccard (incidence-based) indices [20]. Then, we used the SIMPER method for assessing which species has contributed the most to the observed differences between groups of samples [21]. To assess hourly variations in the mosquito community composition, we used the Dominance (1-Simpson) index, in which values closer to 0 indicate species are equally present and values closer to 1 indicate the presence of highly dominant species. Analyses were done using PAST v4 [22].

Stochastic compartmental model

We used a stochastic compartmental model to simulate the population dynamics of mosquito species to determine the effectiveness of adulticide applications at different times of the day.

We selected the most epidemiologically important mosquito vector species that were collected in Brownsville, Texas: *Aedes aegypti, Aedes albopictus, Culex coronator, Culex nigripalpus,* and *Culex quinquefasciatus*; and in Miami-Dade County, Florida: *Ae. aegypti, Cx. coronator, Cx. nigripalpus,* and *Cx. quinquefasciatus* [23–29].

The mathematical model for the dynamics of the vector mosquito populations considers four developmental stages: eggs (E), larvae (L), pupae (P), and the adult female (A) population.

In our mathematical model, we assume that under suitable conditions, eggs hatch into larvae at rate d_E and some suffer natural mortality at rate m_E . Upon feeding on micro-organisms, and after moulting three times, the larvae then develop into pupae at rate d_L . In our model, we consider the carrying capacity K of the larval habitats to limit the maximum number of eggs that can be allowed to develop. As such, we assume no further competition for food among the larvae; hence they only suffer natural mortality at rate m_L . Pupae develop to become adult mosquitoes at rate d_P and die at rate m_P . We considered a 1:1 sex ratio among adult mosquitoes [30]. In the adult stage, we consider only female adult mosquitoes as males are not epidemiologically relevant. We considered the gonotrophic cycles to occur at rate d_A and n_E represents the average numbers of eggs laid per each oviposition. Finally, adult mosquitoes die at rate m_A . All model parameters are species-dependent.

The described process can mathematically be represented by the following system of equations:

$$\dot{E} = n_E d_A \left(1 - rac{E}{K}
ight) A - m_E E - d_E E$$

 $\dot{L} = d_E E - m_L L - d_L L$
 $\dot{P} = d_L L - m_P P - d_P P$
 $\dot{A} = rac{1}{2} d_P P - m_A A$

In our mathematical model, transitioning from one life stage to another is a stochastic process modelled as binomial transitions. We utilize a discrete-time stochastic version of our mathematical model, with time a step of 0.1 days. The model was implemented in R.

Development and mortality rates

All the development and mortality rates used in our model are temperature- and speciesdependent and we used a constant temperature of 25 degrees Celsius. The use of a constant temperature guarantees that the model can reach a stable equilibrium for each species, which makes our analysis independent from the initial condition. The development and mortality rates for *Ae. albopictus* were obtained from Delatte et al. [30] and implemented as in Poletti et al. [31]. The larvae and pupae development ratios for *Ae. aegypti* and *Cx. quinquefasciatus* were obtained from Grech et al. [32]. The egg hatching and the gonotrophic times for the *Ae. aegypti* were obtained from Farnesi et al. [33] and Garcia-Rejon et al. [34] respectively. Additionally, the egg hatching and the gonotrophic cycle for the *Cx. quinquefasciatus* were taken from Handel [35] and Laporta & Sallum [36] respectively. The mortality rates for eggs, larvae, pupae, and adult *Cx. quinquefasciatus* mosquitoes were obtained from Ewing et al. [37] and Watanabe et al. [38], and Farnesi et al. [33], Yang et al. [39] and Yang et al. [40] for *Ae. aegypti*, respectively. Model parameters are reported in S5 Table.

Simulation model of adulticide spraying

Four frequencies of applications of adulticides at a given time of the day are considered: i) twice a day (within the same hour, resembling real-world mosquito control operations) for 5 consecutive days, ii) once a month for two months, iii) once a week for two months, and iv) once every other week for two months. We consider adulticide spraying to affect only adult mosquitoes that are active at the time of spraying. As such, for each study location, hour of the day, and mosquito species, we estimated the fraction of active mosquitoes as: $r_{ii}(t) = C_{ii}(t) / \max_i [C_{ii}(t)]$, where $C_{ii}(t)$ represents the estimated number of active adult mosquitoes in location $i = \{Brownsville, Miami\}$ of species j, at time of the day t. To account for the uncertainty in the number of captured mosquitoes at each time of the day, $C_{ii}(t)$ has been sampled from a Poisson distribution of mean $C_{ii}(t)$, representing the observed number of captured mosquitos in location *i*, of species *j*, at time of the day *t*. Upon spraying, the adult mosquito population is instantly reduced by $\sigma r_{ii}(t)$, where σ represents the spraying efficacy. Spraying efficacy is set at 50% in the main analysis, while 30% and 70% are investigated in the supplementary analyses. To estimate the uncertainty in the effectiveness of the interventions, for each location, mosquito vector species, and time of the adulticide spraying, we performed 1,000 simulations, one for each sampled value of $C_{ii}(t)$. To estimate the effect of the application of adulticides, we run the model until equilibrium is reached and only then we simulate interventions.

Results

Descriptive analysis

A total of 14,502 mosquitoes comprising 17 species were collected in Brownsville and 10,948 mosquitoes comprising 19 species were collected in Miami-Dade County. *Aedes aegypti* was the most common mosquito species collected every hour of the day in both Brownsville and Miami-Dade County. It was also the most abundant species, totaling 7,024 and 4,444 collected specimens in Brownsville and Miami-Dade County, respectively. *Culex quinquefasciatus* was the second most abundant species in Brownsville with 4,473 specimens; however, it was only collected from 6 PM to 7 AM. On the other hand, *Cx. quinquefasciatus* was not abundant in Miami-Dade County, it was the fifth most abundant species collected (237 specimens), mostly between dusk and dawn (S1 and S2 Tables).

Species richness

Species richness varied greatly according to the time of the day with most species being more common between dusk and dawn. In Brownsville, the highest number of species was 15 at 10 PM and the lowest was 3 at 11 AM, 1 PM, and 2 PM. In Miami-Dade County, the highest number of species collected was 14 at 9 PM and the lowest was 4 species at 9 AM (Figs 1 and S1).

The PERMANOVA analyses using the Bray-Curtis and Jaccard indices yielded significant results for the diel activity patterns of mosquito species in both Brownsville and Miami-Dade County. The subsequent SIMPER analyses resulted in distinct scenarios. In Brownsville, *Psorophora cyanescens, Culex erraticus, Aedes sollicitans*, and *Psorophora ciliata* contributed approximately 75% of all the variation in the mosquito community composition. Important mosquito vector species such as *Cx. nigripalpus* (7.6%), *Cx. quinquefasciatus* (<0.001%), *Ae. aegypti* (<0.001%), and *Cx. coronator* (<0.001%) did not contribute substantially to the variation (Table 1). A different scenario was observed in Miami-Dade County, with *Ae. aegypti, Aedes taeniorhynchus*, and *Wyeomyia vanduzeei* accounting for more than 80% of all the variation in



Fig 1. Diel activity patterns of mosquito vector populations. (A) Total number of female adult mosquito specimens collected over the entire study period by hour of collection in Brownsville, Texas. The figure reports vector mosquito species only; <u>S1 Fig</u> shows all the collected species. The Dominance Index is shown in black (mean and 95% CI). **(B)** Same as A, but for Miami-Dade County, Florida.

https://doi.org/10.1371/journal.pntd.0011074.g001

the mosquito community composition. Vector species such as *Ae. albopictus*, *Cx. quinquefasciatus*, *Cx. coronator*, and *Cx. nigripalpus* did not contribute substantially to the variation in the community composition (S3 and S4 Tables).

Effectiveness of adulticide spraying at different hours of the day

We leveraged a model of the mosquito population dynamics to test the effectiveness of adulticide interventions at different times of the day (S2 Fig). Specifically, we considered two scenarios representing two distinct epidemiological situations. Scenario one is during an ongoing arboviral outbreak in which the mosquitoes vector populations must be reduced quickly to curtail the epidemic. Scenario two is an effort to decrease the likelihood of an arboviral outbreak by reducing the sizes of the mosquito vector populations. In scenario two, adulticide applications may be especially important in low-resource settings or in specific situations such as before outdoor sporting events or during holidays in touristic areas when source reduction or the application of larvicides are not practical. Therefore, two different adulticide application

PERMANOVA	Brownsville		Miami-Dade	
	Bray-Curtis	Jaccard	Bray-Curtis	Jaccard
Permutations	9999	9999	9999	9999
Total sum of squares	2.644	3.313	4.78E+06	2.739
Within-group sum of squares	1.024	1.341	2.20E+06	1.119
F	5.7	5.293	4.212	5.213
P	0.0004	0.0001	0.0018	0.0001
SIMPER	Average dissimilarity	Contribution %	Average dissimilarity	Contribution %
Aedes aegypti	4.34E-07	1.97E-06	14.97	48.81
Aedes albopictus	2.06E-08	9.33E-08	-	-
Culex coronator	5.73E-08	2.60E-07	0.7431	2.423
Culex nigripalpus	1.696	7.697	0.3822	1.246
Culex quinquefasciatus	6.16E-07	2.79E-06	1.148	3.743

Table 1. Permutational Multivariate Analysis of Variance (PERMANOVA) and SIMPER (Similarity Percentage) analysis of which species contributed the most to the observed differences in Brownsville, Texas, and Miami-Dade County, Florida.

https://doi.org/10.1371/journal.pntd.0011074.t001

regimes to reduce adult mosquito population sizes were evaluated; (i) adulticide applications twice a day for five consecutive days and (ii) adulticide applications once a week for two months. The adulticide treatment regimens were developed to simulate distinct epidemiological scenarios that are commonly faced by local mosquito control authorities. Input from local vector control agencies was considered to define these treatment regimes, but no official recommendation was available. For sensitivity analysis, we considered two alternative frequencies, once a month for two months and once every other week for two months. The main analysis considers the adulticide to be 50% effective; 30% and 70% are used for sensitivity analyses presented in the Supplementary Material (S3 and S4 Figs).

The estimated effectiveness of adulticide spraying to control populations of adult vector mosquito species varied greatly according to the hour of the intervention for both daily and weekly applications. Regarding the spraying of adulticide twice per day for 5 consecutive days, even though *Ae. aegypti* was collected during all 24 hours of the day, it was more abundant in the morning and evening and the maximum effectiveness was obtained between 8 AM and 12 PM and 7 PM and 11 PM in Brownsville and Miami-Dade County, respectively. Our results show that adulticide interventions done during these hours of the day reach more than 80% effectiveness in reducing *Ae. aegypti* populations. The results for *Ae. albopictus* indicate that adulticide interventions in the evening (between 5 PM and 11 PM) were the most effective reaching a maximum 95.9% reduction (95% CI: 95.3 to 96.4%). Similar results were found for *Cx. quinquefasciatus, Cx. coronator,* and *Cx. nigripalpus* in which adulticide applications during the day from 9 AM to 6 PM were mostly ineffective, having a negligible impact on the abundance of mosquitoes from this species. On the other hand, adulticide applications between 6 PM and 8 AM achieved the best outcomes with a maximum of 95.7% reduction (95% CI: 95.0 to 95.9%) in the abundance of these species (Fig 2A and 2B).

Furthermore, the time of the adulticide intervention was associated with the number of days it took for a mosquito population to return to abundance levels prior to the chemical intervention. It would take 43.8 (95% CI: 38.9 to 50.4) days for the *Ae. aegypti* population to return to abundance levels prior to the adulticide intervention at its peak activity hours, and adulticide interventions during hours *Ae. aegypti* is not active lead to a less effective reduction, taking 30.2 (95% CI: 25.2 to 38.3) days to return to the abundance levels prior to the chemical intervention.

Adulticide interventions between 6 PM to 8 AM were the most effective in substantially reducing *Cx. coronator*, *Cx. nigripalpus*, and *Cx. quinquefasciatus* mosquitoes. As a result, it



Fig 2. Effectiveness of adulticide application twice per day for five consecutive days by hour of application. (A) Maximum reduction (%) of the number of mosquitoes following adulticide application by hour of application in Brownsville, Texas. Adulticide was applied twice per day (within the same hour) for five consecutive days. The maximum reduction is obtained after the fifth day of insecticide application; after that, the mosquito population starts to recover. Lines represent mean values, which are obtained by averaging over 1,000 stochastic model realizations. Bar plots show mean and 95%CI of the same metric for adulticide application at 2 PM and 9 PM. Spraying efficacy is set at 50%. (B) Same as A, but for Miami-Dade County, Florida. (C) Number of days after the end of the adulticide application it takes for the mosquito population to recover to the pre-intervention level by hour of adulticide application. Adulticide was applied twice per day (within the same hour) for five consecutive days. Lines represent mean values, which are obtained by averaging over 1,000 stochastic model realizations. Bar plots show mean and 95%CI of the same metric for adulticide application at 2 PM and 9 PM. Intervention efficacy is set at 50%. (D) Same as C, but for Miami-Dade County, Florida.

https://doi.org/10.1371/journal.pntd.0011074.g002

would take 42.1 (95% CI: 34.3 to 54.2) days for the *Cx. quinquefasciatus* population to reach the same level prior to the adulticide intervention, 34.5 (95% CI: 22.1 to 52.8) days for *Cx. coronator*, and 39.5 (95% CI: 28.8 to 63.7) days for *Cx. nigripalpus*. However, adulticide interventions were much less effective and often generated negligible results in reducing *Culex* populations if carried out between 8 AM and 6 PM, especially in Brownsville (Figs <u>2C, 2D</u> and <u>S5</u>).

The results for adulticide application once per week for two months indicate that mosquito populations can be reduced by up to 40.7% when the intervention is carried out while mosquitoes are more active. The *Ae. aegypti* population from Brownsville was substantially reduced when the adulticide application was carried out between 8 AM and 12 PM and reached optimum results at 8 PM and 9 PM, whereas in Miami-Dade County adulticide applications in the morning were not as effective as in Brownsville, and the most reduction of 39.8% (95% CI: 39.6 to 40.1%) was obtained in the evening at 9 PM. The results for *Ae. albopictus* in Brownsville were similar to those for *Ae. aegypti* in Miami-Dade County. The most effective hour to control *Ae. albopictus* populations was between 7 PM and 10 PM, reaching a 40.7% (95% CI: 39.6 to 42.0%) reduction at 9 PM. Adulticide application at any other hour of the day resulted in a maximum of 15.9% reduction (95% CI: 9.9 to 22.8%).

The results for *Cx. quinquefasciatus* and *Cx. coronator* were similar in both Brownsville and Miami-Dade County. In Brownsville, the maximum reduction was between 7 PM and 8 AM (36.7% reduction at 4 AM; 95% CI: 36.2 to 37.3%) for *Cx. quinquefasciatus* and 9 PM and 7 AM (33.2% reduction at 3 AM; 95% CI: 23.7 to 37.6%) for *Cx. coronator*. In Miami-Dade County, the maximum reduction was between 9 PM and 8 AM (31.7% at 5 AM; 95% CI: 20.2 to 38.1%) for *Cx. coronator* and between 9 PM and 9 AM (33.3% reduction with 95% CI: 22.6 to 37.8% at 3 AM; 31.1% reduction with 95% CI: 20.3 to 37.5% at 7 AM) for *Cx.*





https://doi.org/10.1371/journal.pntd.0011074.g003

quinquefasciatus. The most effective hour of the day to control *Cx. nigripalpus* was found to be between 9 PM to 11 PM (29.9% reduction at 9 PM; 95% CI: 18.9 to 37.6%) and between 12 AM to 2 AM in Brownsville. In Miami-Dade County, the most effective hour of the day to control *Cx. nigripalpus* was between 8 PM and 2 AM (31.0% reduction with 95% CI: 15.4 to 38.5%) and at 9 PM (31.9% reduction with 95% CI: 19.4 to 39.1%) (Figs 3 and S6).

In terms of timing of adulticide application, results assuming a different frequency of the intervention and those assuming a different efficacy of the adulticide are consistent with those reported in the main analysis, albeit with lower absolute reductions (S3 and S4 Fig).

Discussion

Scientific evidence on data on diel activity patterns of different vector mosquito species in different areas is lacking. This is because diel activity patterns are not routinely investigated during the development and implementation of mosquito control strategies. The current mosquito control strategies try to target peak activity times for adulticide applications, but rarely account for local and geographic variations in activity patterns of the target vector populations, thus, failing to reach the maximum effectiveness in controlling adult populations of the vector species. Our results showed that the mosquito community composition and abundance varied significantly throughout the day in both Brownsville, Texas, and Miami-Dade County, Florida. However, in Brownsville, epidemiologically important mosquito species did not significantly contribute to the hourly variation in the mosquito community composition, whereas in Miami-Dade County, Ae. aegypti, was the most dominant species and accounted for approximately 50% of the variation in the mosquito community composition. Even though fewer species were collected in Brownsville and Ae. aegypti was the dominant species from 8 AM to 5 PM, other epidemiologically relevant mosquito species (e.g., Cx. quinquefasciatus and Ae. albopictus) were abundantly found between dusk and dawn. On the other hand, Ae. aegypti was the most abundant species collected in Miami-Dade County at all hours of the day, and despite the higher species richness when compared to Brownsville, the bimodal diel activity pattern of Ae. aegypti was a major driver for the variation found in the mosquito community composition.

Considering the natural variations in the diel activity patterns of mosquito vector species present in Brownsville and Miami-Dade County, we modeled adulticide applications at every hour of the day. Our results indicate that the effectiveness of adulticide applications varied greatly according to the hour of the day and most effective at the peak activity time of the target

mosquito vector species. Should the intervention be aimed at reducing mosquito populations as much as possible in the shortest amount of time, as would be the case when an arbovirus outbreak is ongoing, we simulated a scenario where the adulticide is sprayed twice per day for 5 consecutive days. In the case of Ae. aegypti (the primary vector of Zika, dengue, and chikungunya viruses in the study locations) in both Brownsville and Miami-Dade County, targeting the adulticide application when they are most active, 8 AM and 8 PM, resulted in the reduction of approximately 90% of the population. However, adulticide applications targeting Ae. aegypti in the early morning (between 1 AM and 7 AM) reduced only around 40% of the population. By targeting Ae. aegypti at its peak activity hour, it would take approximately 60 days for the population to reach the same abundance as prior to the adulticide intervention. However, targeting Ae. aegypti in the early morning would result in a less effective mosquito control intervention with the population reaching the same numbers as prior to the adulticide intervention after 40 days. Our results show that a simple change in the adulticide spraying from 7 AM to 8 AM would substantially impact the effectiveness of the mosquito control intervention. Spraying adulticide twice per day for 5 consecutive days yielded even more pronounced results for *Cx. coronator*, *Cx. nigripalpus*, and *Cx. quinquefasciatus*. Our results show that spraying adulticide between 9 AM and 6 PM achieved a negligible reduction in the *Culex* species. On the other hand, adulticide spraying at 9 PM led to a 90% reduction in the number of mosquitoes. As a result, just one hour difference in targeting Cx. quinquefasciatus can lead to a 70% reduction at 7 PM as opposed to a negligible reduction at 6 PM. Similar results were found for Cx. coronator and Cx. nigripalpus, in which spraying at 10 PM led to a 90% reduction but at 8 PM the reduction would have been negligible. By targeting the *Culex* species at their peak activity time, their populations would need approximately 50 days to rebound to the levels prior to the adulticide spraying in contrast with no reduction if the intervention would have been done just a few hours earlier. From an epidemiological standpoint, it is important to note that a 90% reduction in the vector population over a time span of 5 days (i.e., less than a generation time of Zika, dengue, and chikungunya [41-43]) would have a substantial effect on the control of an arbovirus outbreak, causing reproduction numbers of up to 10 to fall below the epidemic threshold of 1. In fact, the reproduction number (i.e., the number of secondary infections generated by an infective host through vector bites in a fully susceptible host population) for a mosquito-borne infectious disease is nearly directly proportional to the number of adult female vectors per human host. Further modelling analyses simulating the pathogen transmission dynamics would be warranted to quantify the non-linear effect of adulticide application on key epidemiological metrics such as the size and duration of an epidemic outbreak, and disease burden. It is also important to stress that incorporating mosquito behavior patterns other than host-seeking (e.g., sugar seeking) could provide more reliable estimations of the effect of adulticide application.

When interventions are aimed at keeping mosquito vector populations to acceptable levels, for instance, to decrease the likelihood of an arbovirus outbreak, we simulated spraying adulticide in more sparse intervals for two months (e.g., once per week, once every other week, and once per month). Also, in this case, the effectiveness of the intervention was positively associated with the peak activity time of each target mosquito vector species. For instance, for *Ae. aegypti*, the estimated reduction ranged from 20.1% (95% CI: 19.7 to 20.6%) and 20.1% (95% CI: 19.5 to 20.4%) when spraying is performed at 9 PM in both Miami-Dade County and Brownsville to 4.3% (95% CI: 3.1 to 5.3%) and 2.8% (95% CI: 1.9 to 3.8%) when spraying is performed at 2 PM.

Our modeling analysis considers the worst-case scenario in which the temperature is kept constant at 25 degrees Celsius–an optimal temperature for mosquito development and proliferation [32]. As such, our estimates can be considered as a lower bound for the effectiveness of

the intervention and are not able to capture the natural seasonal variation in mosquito relative abundance [31,44] and diel activity patterns [17]. Further analyses are needed to provide operational guidance for vector control operations accounting for seasonal variations. A limitation of our analysis is due to the unavailability of the development and mortality ratios for Cx. coronator and Cx. nigripalpus in the scientific literature, we have used the parameters that were available for *Cx. quinquefasciatus*. The mosquito species of the *Culex* genus are organized into 26 subgenera, and both Cx. coronator and Cx. nigripalpus share the same subgeneric classification as *Cx. quinquefasciatus*; they all belong to the subgenus *Culex* [45]. Therefore, even though these are different species they share similar biological traits and the parameters of Cx. quinquefasciatus are reasonable approximations for Cx. coronator and Cx. nigripalpus (S5 Table). Another limitation is that the model considers a single mosquito population at the time, thus ignoring possible competition between species. However, we do not expect the competition for resources to be an important driver for the proliferation of the species selected for this study as they exploit different resources in the urban environment and have distinct population dynamics [46,47]. Finally, the model summarizes into a single parameter all of the complexities determining the carrying capacity of the environment, such as larval habitat availability, possible drying out of larval habitats, immature mosquitoes could be washed out from larval habitats, larval habitat condition (e.g., tires can insulate immature mosquitoes from the elements and have low species richness, natural larval habitats may have predators as well as other species competing for resources), levels of organic material in larval habitats may follow seasonal trends. In sum, there are multiple factors determining the carrying capacity of the environment. This is captured in our modeling work by considering an overall carrying capacity that is included in the equation describing the dynamics of eggs. Therefore, our model neither has the resolution nor the aim of investigating factors determining the carrying capacity of the environment.

Adulticide spraying during the day would potentially increase the risk of exposure of the general public to adulticides and should be considered in the development of mosquito control strategies [48]. Furthermore, different propellant mechanisms (e.g., Buffalo Turbine, Grizzly ULV Sprayer, airplane, etc.) may have different levels of effectiveness due to intrinsic limitations in their operation mechanisms, droplet size, as well as due to different weather conditions (e.g., air temperature and wind speed), presence of vegetation, and other factors present in the target area.

Despite being one of the most important aspects for the development of effective mosquito control strategies based on adulticide spraying and assessing the exposure risk to infective bites, the only recent studies focusing on the diel activity patterns of *Ae. aegypti* in the contiguous United States were done in St. Augustine, Florida, in 2017 by Smith et. al. [49] and in 2020 in Miami-Dade County, Florida, and Brownsville, Texas, by Mutebi et al. [17]. Both studies show a clear bimodal pattern of peak activity of *Ae. aegypti* in the morning and evening.

Adulticides are commonly used in the United States during emergency situations to control and prevent arbovirus outbreaks. For instance, they were widely used during the Zika virus outbreak in Miami-Dade County, Florida [3,50,51]. Our results have great implications to improve the effectiveness of mosquito control strategies. Indeed, diel activity patterns of vector mosquito species should be considered when developing future mosquito control strategies.

In case of an arbovirus outbreak, in addition to adulticide application, it is common practice, as well as recommended by CDC guidelines for surveillance, prevention, and control of West Nile virus [52,53], to establish a perimeter and remove potential aquatic habitats, spray larvicide and adulticide at the point source location of the infection and patient's home address in response to imported and locally transmitted arboviral infections [46,54]. The use of adulticide disregarding the diel activity of the target mosquito species can lead not only to ineffective control interventions but, more importantly, the indiscriminate use of adulticide, even during emergency situations, can have deleterious results by increasing the levels of adulticide resistance of the target mosquito population which may lead to decreased effectiveness of adulticide(s) [15,55,56]. Targeting adult mosquito vector species with adulticide spraying-based control strategies during their peak activity will not only result in more effective control strategies but will also result in the need to spray adulticides fewer times, potentially reducing the development of adulticide resistance by the target mosquito species and the exposure of non-target species.

Conclusion

This study serves as a cornerstone for future studies to improve the effectiveness of mosquito control strategies that employ adulticide applications. Our results point to a well-defined diel activity pattern in the peak activity of mosquito species reaching maximum relative abundance at specific times in both Brownsville, Texas, and Miami-Dade County, Florida. Furthermore, the effectiveness of simulated adulticide applications varied greatly according to the hour of the day with 9 PM being the most effective time of the day to control populations of all vector mosquito species in Miami-Dade County and Brownsville. Our results indicate that the timing of adulticide spraying interventions should be carefully considered by local authorities based on the ecology of mosquito species in the focus area.

Disclaimer

The information provided, and views expressed in this publication do not necessarily reflect the official position of the Centers for Disease Control and Prevention (CDC).

Supporting information

S1 Table. Female mosquitoes captured by BG-Sentinel 2 traps in Brownsville, Texas. (PDF)

S2 Table. Female mosquitoes captured by BG-Sentinel 2 traps in Miami-Dade, Florida. (PDF)

S3 Table. SIMPER (Similarity Percentage) analysis of which species contributed the most to the observed differences in Brownsville, Texas. (PDF)

S4 Table. SIMPER (Similarity Percentage) analysis of which species contributed the most to the observed differences in Miami-Dade, Florida. (PDF)

S5 Table. Mosquito species development and mortality ratios. (PDF)

S1 Fig. Diel activity patterns of mosquito populations. (A) Total number of collected female adult mosquito specimens collected over the entire study period by hour of collection in Brownsville, Texas. The Dominance Index is shown in black (mean and 95% CI). **(B)** Same as A, but for Miami-Dade County.

(TIF)

S2 Fig. Simulated temporal dynamics considering adulticide application. (A) Simulate number of adult females of *Ae. aegypti* when considering adulticide application twice a day at 9 PM for five consecutive days in Brownsville, Texas. Insecticide efficacy is set at 50%. **(B)** Same as A, but adulticide is applied once per week for two months. (TIF)

S3 Fig. Effectiveness of adulticide application for different frequencies of application. (A) Average reduction (%) of the number of mosquitoes following adulticide application by hour of application in Brownsville, Texas. Averages are calculated over the 2-month duration of the intervention. Lines represent mean values and shaded areas represent 95%CI; results are obtained by analyzing 1,000 stochastic model realizations. Insecticide efficacy is set at 50%. (**B**) Same as A, but for Miami-Dade County, Florida. (TIFF)

S4 Fig. Effectiveness of adulticide application once per week for two months for different insecticide efficacies. (A) Average reduction (%) of the number of mosquitoes following adulticide application by hour of application in Brownsville, Texas. Adulticide was applied once per week for two months. Averages are calculated over the 2-month duration of the intervention. Lines represent mean values and shaded areas represent 95%CI; results are obtained by analyzing 1,000 stochastic model realizations. (**B**) Same as A, but for Miami-Dade County, Florida.

(TIF)

S5 Fig. Effectiveness of adulticide application twice per day for five consecutive days by hour of application. (A) Number of days after the end of the adulticide application it takes for the mosquito population to recover to the pre-intervention level by hour of adulticide application. Adulticide was applied twice per day (within the same hour) for five consecutive days. Lines represent mean values and shaded areas represent 95%CI; results are obtained by analyzing 1,000 stochastic model realizations. Insecticide efficacy is set at 50%. **(B)** Same as A, but for Miami-Dade County, Florida.



S6 Fig. Effectiveness of adulticide application once per week for two months by hour of application. (A) Average reduction (%) of the number of mosquitoes following adulticide application by hour of application in Brownsville, Texas. Adulticide was applied once per week for two months. Averages are calculated over the 2-month duration of the intervention. Lines represent mean values and shaded areas represent 95%CI; results are obtained by analyzing 1,000 stochastic model realizations. Insecticide efficacy is set at 50%. **(B)** Same as A, but for Miami-Dade County, Florida.

(TIF)

Author Contributions

Conceptualization: André B. B. Wilke, John-Paul Mutebi, Marco Ajelli.

Data curation: Chalmers Vasquez, Maday Moreno, William D. Petrie, Art Rodriguez.

Formal analysis: André B. B. Wilke, Adequate Mhlanga, Allisandra G. Kummer.

Investigation: André B. B. Wilke.

Methodology: André B. B. Wilke.

Supervision: John-Paul Mutebi, Marco Ajelli.

Writing – original draft: André B. B. Wilke.

Writing – review & editing: Adequate Mhlanga, Allisandra G. Kummer, Chalmers Vasquez, Maday Moreno, William D. Petrie, Art Rodriguez, Christopher Vitek, Gabriel L. Hamer, John-Paul Mutebi, Marco Ajelli.

References

- 1. Monaghan AJ, Eisen RJ, Eisen L, McAllister J, Savage HM, Mutebi JP, et al. Consensus and uncertainty in the geographic range of *Aedes aegypti* and *Aedes albopictus* in the contiguous United States: Multi-model assessment and synthesis. PLoS Comput Biol. 2019; 15: 1–19.
- LaDeau SL, Leisnham PT, Biehler D, Bodner D. Higher mosquito production in low-income neighborhoods of Baltimore and Washington, DC: Understanding ecological drivers and mosquito-borne disease risk in temperate cities. Int J Environ Res Public Health. 2013; 10: 1505–1526. <u>https://doi.org/10.3390/</u> ijerph10041505 PMID: 23583963
- Likos A, Griffin I, Bingham AM, Stanek D, Fischer M, White S, et al. Local Mosquito-borne transmission of Zika virus—Miami-Dade and Broward Counties, Florida, June–August 2016. MMWR Morb Mortal Wkly Rep. 2016; 65: 1032–1038. https://doi.org/10.15585/mmwr.mm6538e1 PMID: 27684886
- Rosenberg R, Lindsey NP, Fischer M, Gregory CJ, Hinckley AF, Mead PS, et al. Vital Signs: Trends in reported vectorborne disease Cases—United States and Territories, 2004–2016. MMWR Morb Mortal Wkly Rep. 2018; 67: 496–501. https://doi.org/10.15585/mmwr.mm6717e1 PMID: 29723166
- Centers for Disease Control and Prevention. Arbonet. Available from at: <u>https://wwwn.cdc.gov/arbonet/maps/ADB_Diseases_Map/index.html</u>.
- Mutebi JP, Hughes HR, Burkhalter KL, Kothera L, Vasquez C, Kenney JL. Zika virus MB16-23 in mosquitoes, Miami-Dade County, Florida, USA, 2016. Emerg Infect Dis. 2018; 24: 808–810. https://doi.org/ 10.3201/eid2404.171919 PMID: 29400646
- Florida Department of Health. Health Officials Issue Mosquito-Borne Illness Advisory. 2016; 4–5. Available from: http://monroe.floridahealth.gov/newsroom/2016/06/160601-Dengue.html.
- Coatsworth H, Lippi CA, Vasquez C, Ayers JB, Stephenson CJ, Waits C, et al. A molecular surveillance-guided vector control response to concurrent dengue and West Nile virus outbreaks in a COVID-19 hotspot of Florida. The Lancet Regional Health—Americas. 2022; 11: 100231.
- Ajelli M, Moise IK, Hutchings TCSG, Brown SC, Kumar N, Johnson NF, et al. Host outdoor exposure variability affects the transmission and spread of Zika virus: Insights for epidemic control. PLoS Negl Trop Dis. 2017; 11: e0005851. https://doi.org/10.1371/journal.pntd.0005851 PMID: 28910292
- Zohdy S, Schwartz TS, Oaks JR. The coevolution effect as a driver of spillover. Trends Parasitol. 2019; 35: 399–408. https://doi.org/10.1016/j.pt.2019.03.010 PMID: 31053334
- Lizzi KM, Qualls WA, Brown SC, Beier JC. Expanding Integrated Vector Management to promote healthy environments. Trends Parasitol. 2014; 30: 394–400. <u>https://doi.org/10.1016/j.pt.2014.06.001</u> PMID: 25028090
- Pagel J, Schurr FM. Forecasting species ranges by statistical estimation of ecological niches and spatial population dynamics. Glob. Ecol. Biogeogr. 2012; 21: 293–304.
- **13.** Clavel J, Julliard R, Devictor V. Worldwide decline of specialist species: Toward a global functional homogenization? Front Ecol Environ. 2011; 9: 222–228.
- Wilke ABB, Vasquez C, Carvajal A, Medina J, Chase C, Cardenas G, et al. Proliferation of *Aedes* aegypti in urban environments mediated by the availability of key aquatic habitats. Sci Rep. 2020; 10: 12925.
- Mundis SJ, Estep AS, Waits CM, Ryan SJ. Spatial variation in the frequency of knockdown resistance genotypes in Florida Aedes aegypti populations. Parasit Vectors. 2020; 13: 241.
- Wilke ABB, Vasquez C, Carvajal A, Ramirez M, Cardenas G, Petrie WD, et al. Effectiveness of adulticide and larvicide in controlling high densities of *Aedes aegypti* in urban environments. PLoS One. 2021; 16: e0246046.
- Mutebi J-P, Wilke ABB, Ostrum E, Vasquez C, Cardenas G, Carvajal A, et al. Diel activity patterns of two distinct populations of *Aedes aegypti* in Miami, FL and Brownsville, TX. Sci Rep. 2022; 12: 5315.
- Wilke ABB, Carvajal A, Medina J, Anderson M, Nieves VJ, Ramirez M, et al. Assessment of the effectiveness of BG-Sentinel traps baited with CO2 and BG-Lure for the surveillance of vector mosquitoes in Miami-Dade County, Florida. PLoS One. 2019; 14: e0212688. <u>https://doi.org/10.1371/journal.pone.</u> 0212688 PMID: 30794670
- Darsie RF Jr., Morris CD. Keys to the Adult Females and Fourth Instar Larvae of the Mosquitoes of Florida (Diptera, Culicidae). 1st ed. Technical Bulletin of the Florida Mosquito Control Association; 2000.
- **20.** Anderson MJ. Permutational multivariate analysis of variance (PERMANOVA). Wiley statsref: statistics reference online. 2014; 14: 1–5.
- Clarke KR. Non-parametric multivariate analyses of changes in community structure. Austral Ecol. 1993; 18: 117–143.
- 22. Hammer Ø, Harper DATT, Ryan PD. PAST: Paleontological Statistics Software Package for Education and Data Analysis. Palaeontologia Electronica. 2001; 4: 9.

- Alto BW, Connelly CR, O'Meara GF, Hickman D, Karr N. Reproductive biology and susceptibility of florida *Culex coronator* to Infection with West Nile virus. Vector Borne Zoonotic Dis. 2014; 14: 606–614.
- Sardelis MR, Turell MJ, Dohm DJ, O'guinn ML. Vector competence of selected North American Culex and Coquillettidia mosquitoes for West Nile virus. Emerg Infect Dis. 2001; 7: 1018–22
- 25. Gutiérrez-Bugallo G, Boullis A, Martinez Y, Hery L, Rodríguez M, Bisset JA, et al. Vector competence of *Aedes aegypti* from Havana, Cuba, for dengue virus type 1, chikungunya, and Zika viruses. PLoS Negl Trop Dis. 2020; 14: e0008941.
- O'connorid O, Calvez E, Inizan C, Pocquet N, Richard V, Dupont-Rouzeyrol M. Vector competence of *Aedes aegypti* from New Caledonia for the four recent circulating dengue virus serotypes. PLoS Negl Trop Dis. 2020; 14: 1–13.
- 27. Dieme C, Ciota AT, Kramer LD. Transmission potential of Mayaro virus by *Aedes albopictus*, and *Anopheles quadrimaculatus* from the USA. Parasit Vectors. 2020; 13: 613.
- Gloria-Soria A, Payne AF, Bialosuknia SM, Stout J, Mathias N, Eastwood G, et al. Vector competence of *Aedes albopictus* populations from the Northeastern United States for chikungunya, dengue, and Zika Viruses. Am J Trop Med Hyg. 2021; 104: 1123–1130.
- Honório NA, Wiggins K, Câmara DCP, Eastmond B, Alto BW. Chikungunya virus vector competency of Brazilian and Florida mosquito vectors. PLoS Negl Trop Dis. 2018; 12: 1–16. <u>https://doi.org/10.1371/</u> journal.pntd.0006521 PMID: 29879121
- Delatte H, Gimonneau G, Triboire A. Influence of temperature on immature development, survival, longevity, fecundity, and gonotrophic cycles of *Aedes albopictus*, vector of chikungunya and dengue in the Indian Ocean. J Med Entomol. 2009; 46: 33–41.
- Poletti P, Messeri G, Ajelli M, Vallorani R, Rizzo C, Merler S. Transmission potential of chikungunya virus and control measures: The case of Italy. PLoS One. 2011; 6: e18860. <u>https://doi.org/10.1371/</u> journal.pone.0018860 PMID: 21559329
- Grech MG, Sartor PD, Almirón WR, Ludueña-Almeida FF. Effect of temperature on life history traits during immature development of *Aedes aegypti* and *Culex quinquefasciatus* (Diptera: Culicidae) from Córdoba city, Argentina. Acta Trop. 2015; 146: 1–6.
- Cristina Farnesi L, Jesus Martins A, Valle D, Lazzaro Rezende G. Embryonic development of Aedes aegypti (Diptera: Culicidae): Influence of different constant temperatures. Mem Inst Oswaldo Cruz. 2009; 104: 124–126.
- **34.** Garcia-Rejon JE, Ulloa-Garcia A, Cigarroa-Toledo N, Pech-May A, Machain-Williams C, Cetina-Trejo RC, et al. Study of *Aedes aegypti* population with emphasis on the gonotrophic cycle length and identification of arboviruses: Implications for vector management in cemeteries. Rev Inst Med Trop Sao Paulo. 2018; 20: 60:e44.
- van Handel E. Fuel Metabolism of the Mosquito (Culex quinquefasciatus) Embryo. J Insect fhysiol. 1993; 39: 831–833.
- Laporta GZ, Sallum MA. Density and survival rate of *Culex quinquefasciatus* at Parque Ecológico do Tietê, São Paulo, Brazil. J Am Mosq Control Assoc. 2008; 24: 21–7.
- Ewing DA, Cobbold CA, Purse B v., Nunn MA, White SM. Modelling the effect of temperature on the seasonal population dynamics of temperate mosquitoes. J Theor Biol. 2016; 400: 65–79. <u>https://doi.org/10.1016/i.jtbi.2016.04.008</u> PMID: 27084359
- Watanabe K, Fukui S, Ohta S. Population of the temperate mosquito, *Culex pipiens*, decreases in response to habitat climatological changes in future. Geohealth. 2017; 1: 196–210.
- Yang HM, Macoris M de L da G, Galvani KC, Andrighetti MTM. Follow up estimation of Aedes aegypti entomological parameters and mathematical modellings. BioSystems. 2011; 103: 360–371.
- Yang HM, Macoris MLG, Galvani KC, Andrighetti MTM, Wanderley DMV. Assessing the effects of temperature on the population of *Aedes aegypti*, the vector of dengue. Epidemiol Infect. 2009; 137: 1188– 1202.
- Nishiura H, Kinoshita R, Mizumoto K, Yasuda Y, Nah K. Transmission potential of Zika virus infection in the South Pacific. Int J Infect Dis. 2016; 45: 95–97. <u>https://doi.org/10.1016/j.ijid.2016.02.017</u> PMID: 26923081
- Rocklöv J, Tozan Y. Climate change and the rising infectiousness of dengue. Emerg Top Life Sci. 2019; 3: 133–142. https://doi.org/10.1042/ETLS20180123 PMID: 33523146
- Guzzetta G, Vairo F, Mammone A, Lanini S, Poletti P, Manica M, et al. Spatial modes for transmission of chikungunya virus during a large chikungunya outbreak in Italy: A modeling analysis. BMC Med. 2020; 18: 226. https://doi.org/10.1186/s12916-020-01674-y PMID: 32762750
- 44. Otero M, Solari HG, Schweigmann N. A stochastic population dynamics model for *Aedes aegypti:* Formulation and application to a city with temperate climate. Bull Math Biol. 2006; 68: 1945–1974.

- Harbach RE. Classification within the cosmopolitan genus Culex (Diptera: Culicidae): The foundation for molecular systematics and phylogenetic research. Acta Tropica. 2011. pp. 1–14. https://doi.org/10. 1016/j.actatropica.2011.06.005 PMID: 21741350
- 46. Wilke ABB, Chase C, Vasquez C, Carvajal A, Medina J, Petrie WD, Beier JC. Urbanization creates diverse aquatic habitats for immature mosquitoes in urban areas. Sci Rep. 201; 9: 15335. <u>https://doi.org/10.1038/s41598-019-51787-5 PMID</u>: 31653914
- 47. Wilke ABB, Vasquez C, Medina J, Carvajal A, Petrie W, Beier JC. Community Composition and Yearround Abundance of Vector Species of Mosquitoes make Miami-Dade County, Florida a Receptive Gateway for Arbovirus entry to the United States. Sci Rep. 2019; 9: 8732. <u>https://doi.org/10.1038/ s41598-019-45337-2 PMID: 31217547</u>
- 48. Centers for Disease Control and Prevention. Aerial Spraying. Available from at: <u>https://www.cdc.gov/</u>mosquitoes/mosquito-control/community/aerial-spraying.html.
- 49. Smith M, Dixon D, Bibbs C, Autry D, Xue R de. Diel patterns of Aedes aegypti (Diptera: Culicidae) after resurgence in St. Augustine, Florida as collected by a mechanical rotator trap. J Vector Ecol. 2018; 43: 201–204.
- McAllister JC, Porcelli M, Medina JM, Delorey MJ, Roxanne Connelly C, Godsey MS, et al. Mosquito control activities during local transmission of Zika virus, Miami-Dade County, Florida, USA, 2016. Emerg Infect Dis. 2020; 26: 872–880. https://doi.org/10.3201/eid2605.191606 PMID: 32310079
- **51.** Stoddard PK. Managing *Aedes aegypti* populations in the first Zika transmission zones in the continental United States. Acta Trop. 2018; 187: 108–118.
- Centers for Disease Control and Prevention. West Nile Virus in the United States: Guidelines for Surveillance, Prevention, and Control. Available from at: https://wnvGuidelines.pdf.
- World Health Organization. Global vector control response 2017–2030. Available from at: <u>http://apps.who.int/iris/bitstream/10665/259205/1/9789241512978-eng.pdf?ua=1</u>.
- Centers for Disease Control and Prevention. Guidelines for Arbovirus Surveillance Programs in the United States. 1993. Available from at: https://www.cdc.gov/ncezid/dvbd/adb/resources/arboguid_508. pdf.
- Amorim LB, Helvecio E, de Oliveira CMF, Ayres CFJ. Susceptibility status of *Culex quinquefasciatus* (Diptera: Culicidae) populations to the chemical insecticide Temephos in Pernambuco, Brazil. Pest Manag Sci. 2013; 69: 1307–1314.
- Pocquet N, Milesi P, Makoundou P, Unal S, Zumbo B, Atyame C, et al. Multiple insecticide resistances in the disease vector *Culex p. quinquefasciatus* from Western Indian Ocean. PLoS One. 2013; 8: 1–14.